Author’s response to reviews

Title: Factors associated with infants' and young children's (6-23 months) dietary diversity in Pakistan: Evidence from the Demographic and Health Survey 2012-13

Authors:

Sarosh Iqbal (sarosh.iqbal@gmail.com)
Rubeena Zakar (rubeena499@hotmail.com)
Muhammad Zakar (mzzakir@yahoo.com)
Florian Fischer (f.fischer@uni-bielefeld.de)

Version: 4 Date: 11 Nov 2017

Author’s response to reviews:

Thank you very much for checking the manuscript so carefully. You are absolutely right with both comments (see below). We have changed Table 1 according to your suggestions. In addition, we have corrected the value in Table 5.

---

1. My version of the paper did not show the revised figures in table 1. The % figures should be easily seen by the reader as the number of children reporting each food group is given in the table - the description appears to be correct in the revised text (lines 227-230) but the % figures appear to be incorrect in the table. The numbers of children in each age group should be shown at the top of the column for each age group to make the denominator clear for the reader. Please check the table.

2. Please check Table 5 model B for higher education - I think the CI should be 0.03 without a preceding minus sign?